Author's response to reviews

**Title:** Circulating levels of cell adhesion molecule L1 as a prognostic marker in gastrointestinal stromal tumor patients

**Authors:**

Hilke Zander (hilke.zander@zmnh.uni-hamburg.de)
Tamina Rawnaq (trawnaq@uke.de)
Max Wedemeyer (m.vonwedemeyer@googlemail.com)
Michael Tachezy (mtachezy@uke.de)
Miriam Kunkel (mkunkel@uke.de)
Gerrit Wolters (ge.wolters@uke.de)
Max Bockhorn (m.bockhorn@uke.de)
Melitta Schachner (Schachner@Biology.Rutgers.Edu)
Jacob R Izbicki (izbicki@uke.de)
Jussuf T Kaifi (jkaifi@hmc.psu.edu)

**Version:** 4 **Date:** 11 May 2011

**Author's response to reviews:**

In this last version we made some formatting changes.

Max Bockhorn has been linked to an affiliation.

The reference list was updated, the term 'et al.' has been replaced.

The table title was put above the table.

Sincerely yours

Hilke Zander